Tutorial and homework feedback report

Student: Tutor: Tutorial grade: (“satisfactory” or “needs improvement”)

Tutorial performance

Knowledge acquisition

• Puts in a good effort on all homeworks • Comes to tutorial rested and prepared • Contributes resources and knowledge to group discussion • Learns aggressively • Seeks help when needed

Problem solving

• Adopts reasonable and creative solution strategies • Organizes his/her thinking on paper • Executes technical methods quickly and accurately • Checks work

Technical writing

• Provides clear and concise answers • Reasons in an organized and logical manner • Writes good executive summaries • Includes diagrams and worked examples, where appropriate • Adopts clear mathematical notations • Is neat and uses proper grammar, punctuation, and spelling

Personal and interpersonal development

• Participates punctually, regularly, and cooperatively in lectures and tutorial • Speaks clearly and logically • Listens with openness and empathy to others • Treats others with sensitivity and respect • Adopts a positive team attitude towards problem solving and learning

Comments